Mauna Loa Sol ar Observatory Cbserver’'s Log
Sun May 14 16:33: 08 GVI' 1995
Year: 95 Doy: 134
onserver: yasukawa consol e May 14 16: 29

VWEATHER COMMENT: Sun May 14 16:33:50 GMI 1995
Col d, clear, strong southeast wi nd 20-30 nph.

LOML COWMMENT: Sun May 14 16:34:25 GMI 1995

Low crashed yesterday at 21:29:56, tape unit 1. Renoved

Loo235 fromunit 0 and L0O0236 fromunit 1. Loaded L00237 and L00238.
Reboot ed.

COMMENT: Sun May 14 16:54: 32 GVI 1995

No observations today. Guider still down. Howard could not find
a suitable spare in HAO supply cabinets. He will order a new one
ASAP Monday norning from Vernitron, presuneably. We will await
recei pt of the replacenent notor before repair is conplete and
observati ons can resune.

Sun May 14 16:57:43 GMI 1995: Fil emark



